

G OPEN ACCESS

Citation: Lubinda J, Bi Y, Hamainza B, Haque U, Moore AJ (2021) Modelling of malaria risk, rates, and trends: A spatiotemporal approach for identifying and targeting sub-national areas of high and low burden. PLoS Comput Biol 17(3): e1008669. https://doi.org/10.1371/journal. pcbi.1008669

Editor: Jane M. Heffernan, York University, CANADA

Received: August 23, 2019

Accepted: January 4, 2021

Published: March 1, 2021

Copyright: © 2021 Lubinda et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data used in this study is considered a property of the Republic of Zambia, and therefore subject to ethical and legal restrictions. This means that although it can freely be requested through the Ministry of Health, it cannot be shared without prior approval of the Ministry. Permission to use or access the district level malaria datasets for mortality and morbidity used in the manuscript can be obtained from the National Malaria Elimination Centre (NMEC) through: The Ministry of Health, Ndeke House, RESEARCH ARTICLE

Modelling of malaria risk, rates, and trends: A spatiotemporal approach for identifying and targeting sub-national areas of high and low burden

Jailos Lubinda^{1,2}*, Yaxin Bi³, Busiku Hamainza⁴, Ubydul Haque^{5,6,7}, Adrian J. Moore¹

School of Geography and Environmental Sciences, Ulster University, Coleraine, United Kingdom, 2 School of Computing, Engineering and Intelligent Systems, Ulster University, Londonderry, United Kingdom,
 School of Computing, Ulster University, Newtownabbey, United Kingdom, 4 Ministry of Health, National Malaria Elimination Centre, Lusaka, Zambia, 5 Department of Biostatistics and Epidemiology, University of North Texas Health Science Centre, Fort Worth, Texas, United States of America, 6 Department of Geography, University of Florida, Gainesville, Florida, United States of America, 7 Emerging Pathogens Institute, University of Florida, Gainesville, Florida, United States of America

* lubinda-j@ulster.ac.uk

Abstract

While mortality from malaria continues to decline globally, incidence rates in many countries are rising. Within countries, spatial and temporal patterns of malaria vary across communities due to many different physical and social environmental factors. To identify those areas most suitable for malaria elimination or targeted control interventions, we used Bayesian models to estimate the spatiotemporal variation of malaria risk, rates, and trends to determine areas of high or low malaria burden compared to their geographical neighbours. We present a methodology using Bayesian hierarchical models with a Markov Chain Monte Carlo (MCMC) based inference to fit a generalised linear mixed model with a conditional autoregressive structure. We modelled clusters of similar spatiotemporal trends in malaria risk, using trend functions with constrained shapes and visualised high and low burden districts using a multi-criterion index derived by combining spatiotemporal risk, rates and trends of districts in Zambia. Our results indicate that over 3 million people in Zambia live in highburden districts with either high mortality burden or high incidence burden coupled with an increasing trend over 16 years (2000 to 2015) for all age, under-five and over-five cohorts. Approximately 1.6 million people live in high-incidence burden areas alone. Using our method, we have developed a platform that can enable malaria programs in countries like Zambia to target those high-burden areas with intensive control measures while at the same time pursue malaria elimination efforts in all other areas. Our method enhances conventional approaches and measures to identify those districts which had higher rates and increasing trends and risk. This study provides a method and a means that can help policy makers evaluate intervention impact over time and adopt appropriate geographically targeted strategies that address the issues of both high-burden areas, through intensive control approaches, and low-burden areas, via specific elimination programs.

Haile Selassie Avenue, PO Box, 30205, Lusaka (Zambia) Email: ps@moh.gov.zm Authorisation of the study to use Zambian data was obtained from the Zambia National Health Research Authority, and study protocols/data requested was Reviewed and approved by the Zambian ERES Converge IRB (Ref: 2017-Sept-011). Further verification or inquiry on availability of the dataset used here or other related datasets can also be made from: The National Malaria Elimination Centre Chainama Hills Hospital Grounds Lusaka Zambia Tel/Fax: +26 0211 282427.

Funding: None of the author(s) received any specific funding for this work. The UK Commonwealth Scholarship supported JL while UH was supported in part by the Emerging Pathogens Institute at the University of Florida and the College of Liberal Arts and Sciences, as part of the University of Florida Pre-eminence Initiative.

Competing interests: The authors have declared that no competing interests exist.

Author summary

The WHO Global Technical Strategy for malaria has set an ambitious goal to achieve a global reduction in malaria incidence and mortality rates by 90% and attain elimination in at least 35 countries by the year 2030. Malaria intervention choices and strategies for control and elimination still largely depend on the clear identification and understanding of prevailing variations in local malaria transmission levels. Having reliable and robust methods of selecting and classifying areas as either high or low burden is critical in determining the basis for, and evaluation of, optimal malaria control and elimination strategies at national and sub-national levels. Our study presents a novel statistical method for modelling a combination of malaria risk, rates and trends over time (2000–2015) and space (sub-national level). This approach can be used in malaria endemic countries to help policy makers design, implement and evaluate cost-effective, geographically targeted intervention programmes. These programmes include intensive malaria control approaches in high-burden areas and leveraging malaria elimination strategies in low-burden areas.

Introduction

Malaria transmission trends and risk of infection are usually heterogeneous in time and space. The ability to detect common spatial and temporal variations of malaria burden in subnational settings is of great interest and a considerable challenge to malariologists and public health experts in endemic countries.

The global decline of malaria incidence rates has stalled, or the rate of reduction slowed in some countries, particularly sub-Saharan Africa [1]. The 2017 and 2018 World Malaria Reports highlight this stagnation [1–3] and have led to the World Health Organisation's (WHO) launch of a new country-focused approach known as the "*high-burden to high-impact*" malaria response. They also call for the development of novel methods to address the problem [4,5].

Despite the continued fight against high malaria endemicity for the last half-century, Zambia is among those sub-Saharan countries affected by the reported stagnation in malaria progress [6,7]. With a massive scale-up in interventions [7–10] in the last decade, Zambia achieved considerable progress, resulting in a move away from control targets to elimination aspirations [11]. Zambia embraced the currently renewed global interest for malaria elimination, and strategically positioned itself within a regional and global malaria eradication context.

However, Zambia's geographical location complicates its malaria control status vis-à-vis its elimination aims. For example, the country's northern and south-eastern neighbours (Angola, Congo DR, Tanzania, Malawi, and Mozambique) are often among the WHO's list of *highest-burden countries* [1,2]. In contrast, some of its southern neighbours are regional frontline target countries in the E-2020 malaria elimination programme [12]. Similarly, this northern vs southern epidemiological contrast is manifest sub-nationally as, generally, Zambia's northern regions have high malaria infections while the southern regions experience the opposite [13,14]. Uncertainty in progress both regionally [12] and nationally [15] has not prevented Zambia from moving forward with its aim to eliminate malaria.

In the past, countries have generally embarked on nationwide elimination efforts or intensifying control in low-burden and high burden areas, respectively. Traditionally delineating these areas was logically based on incidence alone. As elimination and control are becoming a focal problem of subnational importance, malaria programs now have to deal with the challenge of accurately delineating areas to pursue elimination and those in which to intensify control strategies, over and above the challenges posed by border areas.

In order to ascertain the robustness of methods used for selecting these areas of a high or low burden to inform optimal control or elimination strategies, and as a measure of progress towards country elimination targets, scholars have started thinking of better or more robust alternatives. Kitojo et al. recently compared multiple data sources such as the use of malaria tests from antenatal care against population-wide prevalence surveys in children under five years of age to evaluate them as a measure for malaria trends and progress towards Tanzania's elimination at subnational levels [16]. Routlege et al. used individual-level malaria cases for geostatistical estimates of spatiotemporal transmission to predict the timeline to elimination or the imminent risk of resurgence in China [17], while Amratia et al. used a combination of serology data, case tracing, and case reports in Haiti [18] to comprehensively capture the transmission landscape.

These studies cite the inadequacy of incidence or prevalence as a single metric, and their methods provide alternatives for multi-metric approaches using multiple data sources besides routine data. However, most endemic countries like Zambia have sufficient routinely collected data but limited population-based survey alternatives. We thus provide an alternative multi-metric approach using a single data source by combining three different measures to better understand and guide the classification of malaria burden and help monitor progress towards malaria control and elimination goals.

While malaria incidence or prevalence rate is a good indicator of how many people need treatment, it only offers a snapshot of infections at a given time point, while missing other important underlying factors such as asymptomatic malaria, and differences in care seeking behaviour. Travel and human movement remain key to malaria elimination, especially in low transmission settings, and any local reductions in prevalence are unlikely to persist if surrounding areas maintain much higher prevalence. Similarly, targeting interventions towards outliers with unusually high levels of malaria burden surrounded by low transmission areas, even after accounting for spatial trends, are likely to be more sustainable in the longer term.

While the logic and justification for targeting high burden areas using incidence alone is sound and deep-rooted in decades of use, the challenge is that low-incidence areas with increasing malaria may still be ignored if incidence is the only defining measure. Ignoring such areas with low but increasing malaria incidence (as a low priority) can compound problems later if these areas progress to moderate or even high incidence status. Considering the trend, however, captures not only the stability in spatial and temporal patterns but also gives an additional perspective in areas where elimination efforts may be ongoing or planned.

In Zambia's approach, elimination is targeted explicitly in subnational areas where the disease exhibits low incidence while control measures are maintained and implemented in the rest of the country [14]. With insufficient levels of funding for malaria control, the *"High burden to high impact"* approach could help reinvigorate the fight against malaria [4] through the more focused and strategic use of evidence-based decision making that can deploy the most effective malaria control tools in areas where they can have maximum impact. The approach presented in this paper supports the identification and targeting of high-burden areas. It also facilitates the optimisation and prioritisation of locally owned country-led health strategies and priorities to achieve their impact maximisation. We add to the literature advocating that, while accepting disease incidence as the primary basis for making decisions on the choice of areas for control or intervention measures, decision making on this basis alone can be optimised and enhanced without incurring any additional data collection costs. We also highlight the ability of our method to define and measure high or low burden areas in line with the *high-burden highimpact* strategy in order to optimise the delivery of control interventions and tools [4]. Identifying the precise quantity and location of the highest-burden areas could help programs focus their limited resources by targeting such areas for further investigations, treatment, prevention efforts, and relevant media campaigns. For example, cost-prohibitive strategies such as mass drug administration (MDA) become feasible for every individual in a small-targeted community hotspot but are not feasible for population-wide application. For instance, Zambia has mostly used targeted indoor residual spray (IRS) to enhance and supplement universal insecticide-treated bed net (ITN) coverage [7]; hence accurate classification is essential in order to ensure the correct application of interventions in true areas of need. Targeting intervention efforts to those places with the highest disease burden relative to surrounding areas is essential because most malaria hot spots are in themselves risks and a source of malaria infections for surrounding areas. Targeting these would help generate a ripple effect that can significantly reduce transmission rates and risk across the recipient areas.

In this study, we investigated the spatiotemporal malaria risk, rate, and trends of all 72 districts in Zambia between 2000 and 2015 using the following process: i) estimate the relative risk and rates of malaria for each district for all ages, under-fives and over-fives, ii) model overall spatial clustering and any related temporal trends and iii) apply a rigorous, but reasonably straightforward, matrix to identify and visualise high burden malaria districts to help inform and support national control and elimination targets. This approach supports and addresses the call for the targeted control or elimination of malaria based on delineated sub-national zones defined by high-burden clusters of risk, rate, and trend.

Methodology

Ethics Statement

The National Health Research Authority authorised the study. Study protocols and data requested was reviewed and approved by Ulster University Research Governance (Ref: 17/ 0049) and the Zambian ERES Converge Institutional Review Board (Ref: 2017-Sept-011).

Study area

Zambia is a landlocked country in South Central Africa, neighbouring eight other malariaendemic countries [7,19], three of which, represent the frontline region-specific Elimination8 (E8) and E-2020 malaria elimination countries [20]. Zambia's geographic location creates a heterogeneous and complex malaria transmission landscape that is suitable for tailored microgeographic intervention approaches.

Box 1. Nomenclature for equations used

ϕ	Random effects	$\mathbf{D} = d_{tj}$	Temporal neighbourhood matrix					
ho s, ho T	Dependence parameters	$\mathcal{T}^2 oldsymbol{Q} (oldsymbol{W}, ho oldsymbol{s})^{-1}$	Variance					
$\overline{{\cal T}_t^2}$	Temporary-varying variance parameter	ρT	Temporal autoregressive parameter					
δ	Overall temporal trend	$f_s(t \gamma_s)$	Spatial trend					
W	Adjacency matrix	k	Spatial unit					
ω_{kj}	Spatial closeness of areal units	ωκ	Binary indicator where $\omega_{\kappa s} = 1$					
ψ	Latent component	λ	Region-wide probability					
t	Timepoint	α	Priori distribution					
https://doi.org/10.1371/journal.pcbi.1008660.t001								

https://doi.org/10.1371/journal.pcbi.1008669.t001

Spatial, population, and malaria data

District populations in the period from 2000 to 2015, were estimated using intercensal and postcensal exponential population growth model information from the Central Statistics Office (CSO) reports from 2000 and 2010 [21]. Post census population estimates and age groups of under-five and over-five-year-olds were obtained from the 2013 CSO report [19]. The derived estimates formed the basis for calculations of malariometric indices such as mortality and morbidity rates by age groups.

The inclusion of two age groups (<5 and ≥ 5) in this study was influenced by two main factors. Firstly, the data was primarily made available in the three age categories of <1, 1–4, and ≥ 5 years. We grouped the first two categories into one 0–4 years to be consistent with the national and global malaria priorities and reporting. Secondly, due to the high susceptibility risk, vulnerability, and severity of exposure to malaria infection or disease among under-five children, it has been a focus and priority in the last decade to track progress in under 5 children malaria mortality and incidence. This has since defined the subsequent focus on the under-fives as reported in all Malaria Indicator Surveys (MIS), Demographic Health Surveys (DHS), and World Malaria Reports (WMRs) [22,23].

We obtained malaria epidemiological data through the Ministry of Health (MoH). Clinical and microscopy-confirmed malaria deaths and cases disaggregated by age groups were reported quarterly before 2008. With the countrywide introduction of rapid diagnostic tests (RDTs) between 2008 and 2011 [9,10,24–27], clinical and confirmed cases were reported separately and monthly [28]. In order to retain the usability of the full dataset from 2000 to 2015, we analysed our data annually and maintained the 72 original districts, using a combination of both confirmed cases and unconfirmed malaria cases. We computed malaria standardised incidence ratios (SIR) per 1000, and standardised mortality ratios (SMR) per 10,000 people using a simple formula: SIR = (Observed Cases/Expected Cases) & SMR = (Observed deaths/Expected deaths).

The data's completeness reporting between 2000 and 2008 was not available at district or health facility level. Instead, national averages were used from WHO's WMR reports which were consistently high over that period, with a median = 87%, mode = 87%, mean = 88.3%, and SD = 2.97%. Completeness data, however, was available at health facility-level and was utilised to more accurately adjust the district-level data between 2009 and 2015. Information on the missingness of data was only available at the district level. Thus, although missingness was dealt with at the district level, it is highly likely that any variations at the facility level will not be detected. However, missingness at district level stood at 3.4% in deaths among those aged 5 years and over, 2.7% in under 5 deaths, and only 0.1% for reported morbidity.

Among the suspected causes of missingness include a lack of adequate training for the person reporting, understaffing, disease burden, levels of education or simply human error in the collection and aggregation of data. An important consideration in this study was the possibility of spatial bias in completeness/missingness of data. We used the Getis-Ord Gi* Spatial Statistic to test for spatial bias in data incompleteness (as a proxy for missingness) and found that incompleteness was random with a Z-score = 0.049 and p-value = 0.96. On this basis, we assumed that any missingness was completely at random (MCAR).

We used Random Forest to impute the 5% of missing values in the data. From missing values among malaria deaths alone, the normalised mean squared error (NMSE) often used to represent error derived from imputing missing values was 0.22 (22%), while it was 0.072 (7%) for missing case values and 0.094 (9%) overall for the whole dataset.

We, however, did not adjust them for confirmation rates by use of Test Positivity Rate (TPR) because TPR was neither available nor collected between 2000 and 2008. In most instances, information on testing was not available at facility or district-level between 2009 and 2015 either.

Spatio-temporal modelling

We used a Conditional Autoregressive (CAR) prior method. The CAR method incorporates spatiotemporal generalised linear mixed models for unique areas with inference in a Bayesian environment using Markov Chain Monte Carlo (MCMC) simulations [29–32]. Our model choice is based on its robustness and capability to estimate the effects of risk factors on response variables such as incidence and mortality [33]. We used the models for identifying clusters of neighbouring districts [34] that display a repeated high risk [35] of malaria compared with other adjacent areas. These models account for spatiotemporal variations within the same environment, mainly when using the CARBayesST R package [33,36,37]. Malaria data counts are observed within districts with an assumption that the data has an independent distribution using a Poison model. The model hierarchy defined and specified within its prior distributions would accommodate for any spatial correlations within the data. (*See S1 Appendix*).

The two main models performed in this study included generalised linear mixed models of various forms. The first generates spatiotemporal patterns in the mean response with a general temporal effect but separate independent spatial effects for each year [38]. This model is defined by Equation (1):

$$\psi = \phi_{kt} + \delta_t,$$

where

$$\phi_{kt} | \phi_{-kt}, \mathbf{W} \sim N \left(\frac{\rho s, \sum_{j=1}^{K} \omega_{kj} \phi_{jt}}{\rho \sum_{j=1}^{K} \omega_{kj} + 1 - \rho}, \frac{\mathcal{T}_{t}^{2}}{\rho \sum_{j=1}^{K} \omega_{kj} + 1 - \rho s} \right),$$

$$\delta_{t} | \boldsymbol{\delta}_{-t}, \mathbf{D} \sim N \left(\frac{\rho \tau, \sum_{j=1}^{N} d_{tj} \delta_{j}}{\rho \tau \sum_{j=1}^{N} d_{tj} + 1 - \rho \tau}, \frac{\mathcal{T}_{t}^{2}}{\rho \tau \sum_{j=1}^{N} d_{tj} + 1 - \rho \tau} \right),$$

$$(1)$$

$$\mathcal{T}_{1}^{2}, \dots, \mathcal{T}_{N}^{2}, \mathcal{T}_{T}^{2}, \sim \text{Inverse} - \text{Gamma}(a, b),$$

 $\rho s, \rho T \sim \text{Uniform}(0, 1).$

We used this model to show the common overall spatial effects for all periods, a common temporal trend, and independent space-time interactions.

The second model is used for districts based on their temporal trends in the risk of malaria infection or death, with trend functions optimised by fixed parametric forms or constrained shapes [35]. We used the model's effects to follow a multivariate autoregressive process with order 1, using the Equation [2]:

$$\psi = \phi_{kt} + \sum_{s=1}^{S} \omega_{\kappa s} f_s(t|\gamma_s), \qquad (2)$$

$$\phi_k | \phi_{-k} \sim N \left(\frac{\rho \sum_{j=1}^K \omega_{kj} \phi_j}{\rho \sum_{j=1}^K \omega_{kj} + 1 - \rho}, \frac{\mathcal{T}^2}{\rho \sum_{j=1}^K \omega_{kj} + 1 - \rho} \right),$$

 $\mathcal{T}^2 \sim \text{Inverse} - \text{Gamma}(a, b),$

 $\rho s, \rho T \sim \text{Uniform}(0, 1).$ $\omega k = (\omega k 1, \dots, \omega k S) \sim \text{Multinomial}(1; \lambda),$ $\lambda = (\lambda_1, \dots, \lambda_s) \sim \text{Dirichlet}(\boldsymbol{\alpha} = (\alpha_1, \dots, \alpha_s)),$ Where $\phi_{-k} = (\phi 1, \dots, \phi_{k-1}, \phi_{k+1}, \dots, \phi K).$

Our model was implemented with 4 MCMC chains and 20000 samples obtained by generating 220 000 samples and removing the first 20 000 as burn-in. We applied thinning on the remaining 200 000 by 10 to reduce the autocorrelation. The outputs from this model include a spatial visualisation (map) with credible intervals, a trend classification probability, a slope of trend change and summaries of the trend outcomes and parameters. However, although we use all these for our interpretation, we do not discuss any of these in the text except the trend visualisation.

Finally, we classified and visualised districts as high-burden or low-burden based on a matrix score using the combined values of relative RIsk, RAtes, and risk Trend (RIRAT) implemented in ArcGIS 10.5 (*See also <u>S1 Appendix</u>*).

Results

The spatiotemporal trend of malaria mortality and incidence rates from 2000 to 2015

Preliminary analysis of results show temporal progress in the reduction of malaria mortality; however, the trend of malaria incidence remains high. Fig 1A shows a significant decline of about 80% in overall malaria mortality from over 11 500 deaths in 2000 down to near 2300 in 2015. Mortality rates among under-five children showed the most significant decline from 28 down to only 3.3 per 10 000 population at a 95% confidence interval, representing a circa 90% decline. Mortality among the over five population also declined from about 5.9 to 0.58 per 10 000 population.

Fig 1B shows a reduction in incidence rates among under-five children from 1457 to 680 (95% CI) per 1000 population with an average reduction of 44 cases annually. Meanwhile, there was a 14% increase in malaria incidence among the over-fives from 224 to 255 per 1000 population (95% CI).

Fig 2 shows the spatiotemporal trends of malaria mortality (2A and 2B) and incidence (2C and 2D) for all ages. Fig 2B and 2D shows temporal trends highlighted by the posterior national median (red) and 95% credible intervals (black) for (i) countrywide mean mortality rates and (ii) the level of spatial standard deviation in mortality and incidence trends. The blue dots are mortality and incidence rates for each district by year. The figures confirm that mortality has declined steadily over the study period with a significant decrease in spatial variance across the 72 districts resulting in a homogenously low risk across the whole country by 2015. In contrast, incidence rates have been unstable with a noticeable increase since 2008, along with an increase in spatial variance across the 72 districts.

Although the change in testing and treatment guidelines around 2008/9, with the introduction of RDTs, could have affected some of the observed declines or no change in trends due to expected decreases in diagnosis of malaria, the observed increases in malaria trends should be deemed independent of this effect because the introduction of RDTs would have eliminated





most of the excess non-malaria fever cases. This is because malaria case counts pre-2008 were primarily overestimations (comprised of true malaria as well as non-malaria fevers) [25]. Besides, the results presented here still conform to those reported by Zambia's microscopy slide-based malaria indicator surveys (MIS) from 2006, 2008, 2010, 2012, and 2015, which show a decline in malaria between 2006 and 2008, but a speedy rise from 2008 to 2015. MISs are independent household surveys undertaken to collect national and subnational data from representative samples of respondents. They assess, among other things, the coverage of key malaria interventions, measure malaria-related burden using prevalence testing of parasites, anaemia, case management, and knowledge and community empowerment among children under five years old.

Results from MISs also confirm the inherent consistency in the trend captured in the routinely collected data. This is further validated by the improving quality of HMIS data observed from the declining portion of unconfirmed malaria reported in the HMIS from 55% in 2011 to only 20% in 2015 (*see S1 Appendix and S3 Fig*). However, these observed spatial variances may be a result of factors such as staggered interventions, especially IRS, which is not applied consistently in specific districts but rather targeted to supplement LLINs in very high transmission areas. This means that the chances of having areas sprayed in one year and not another depending on the preceding year's transmission levels were high. RDT stock-outs [26,39–45]



Fig 2. Box plots vs spatio-temporal trends and deviations in transmission. (a & c) are box plots showing temporal mean and inter-quartile ranges of all age malaria mortality and incidence from 2000–2015. (b&d) show the temporal and spatial standard deviations in mortality and incidence rates (Posterior median (red); 95% credible interval (black); each blue dot representing the mortality/incidence rate for each district and year). Temporal mortality decline is associated with increasing spatial homogeneity (a & b), while the incidence trend is more temporally and spatially heterogeneous over the same period. (c & d).

(recorded at \approx 20% in 2015) or any differences in the adoption of RDT usage by clinicians could just as well cause such spatial variations. We also further observed that the decline of 2008 predates RDTs by 2 years and comes on the backdrop of the removal of health facility user fees that instead should have increased the cases captured and promote a rise rather than a decline.

It would still be fair to assume that RDT adoption or stock could be an issue due to commodity distribution inefficiencies following this RDT implementation, especially for districts further away from the initial national/central hub. Massive stock-outs especially in further off rural districts, were common before the optimisation of the supply chain as summarised in Vledder [45]. Flaws in the medical supply chain management of commodities and equipment, however, have also been acknowledged in many other studies and reports [26,39–44]. Nonetheless, these persistent stock-outs have significantly reduced in number although they may still have random spatio-temporal effects across the period. Despite the challenges highlighted above, it is worth noting that the documented variability in the availability and application of RDTs and control measures across districts was not consistent through time or space, and therefore cannot explain the long term trends discussed here. The spatial patterns for both mortality and incidence rates can be seen in Fig 3A and 3B.



Fig 3. Spatial patterns of malaria mortality and incidence rates, 2000 to 2015. (A) and (B) show the mean spatial pattern of both mortality and incidence rates between the years 2000 and 2015.

Spatial patterns of malaria risk from 2000 to 2015

Fig 4 shows the district level relative standardised mortality risk (SMR) and standardised incidence risk (SIR) for all age, under 5 and over 5 categories. The interpretation of the risk scores is that an SMR/SIR of 1.5 corresponds to a 50% higher risk compared with the countrywide average while an SMR/SIR of 0.9 denotes a 10% lower risk. Based on the calculated SMR, few districts indicate a higher risk of mortality among under-five populations. Notably, some districts in the Eastern and Northern provinces have more than a 250% higher risk of malaria mortality for the under-five age group above the national average, and generally, the Eastern province had the highest risk across the country. The figures also support the temporal trends observed earlier in that the under-fives have a higher risk compared to the over-five age-group. The risk of infections also shows the similar but less extreme variance in spatial patterns with concentrations of low-risk areas in the south and parts of the Central and Northern provinces.

Spatial Clustering of areas exhibiting similar malaria trends

Fig 5 shows the distribution of district clusters exhibiting similar temporal malaria risk trends. Districts were categorised as having either an increasing trend (red), a constant/no change trend (black) or a decreasing trend (blue) with the darker/deeper the shading, the higher the posterior probability for that trend and vice versa. There was very little to no posterior uncertainty in the under-five mortality and incidence risk classifications assigned to each of the three trends (*increasing, constant/no-change*, and *decreasing*). In contrast, minimal uncertainties (*probability* = 0.5-0.75) are visualised in the *increasing* over-five mortality clusters and *no-change* in all-age clusters.

With regard to mortality trends, in those districts with either a constant or decreasing trend, the pattern of change in trend over time has levelled-off and currently remains steady. However, in those districts where the mortality trend has been increasing (i.e. 7% of districts for under-fives and 32% districts for over fives) the pattern of increase during the 16 year study period has been rising. This would indicate that there is a worsening situation in malaria mortality in those areas, creating a real potential to negatively influence national mortality figures if this situation continues (*See S1 Fig*).





Progress in reducing under-five mortality over the 16-year study period is consistent and evident across risk, rates, and trends while incidence across the three age categories is less consistent and more varied. Only 3% (2) of districts showed an increasing trend in under-five mortality while 71% and 26% experienced a decreasing trend or no-change, respectively



Fig 5. Temporal trend of malaria under-five children and over five age group mortality and incidence trends, 2000–2015. The *red* trend shows *increasing*, the blue trend shows *declining*, and the *black* trend shows *constant/no change*. Classifications are based on the maximum posterior probabilities: the *darker/deeper* the shading, the higher the posterior probability for that trend and vice versa. Fig 5(iii) is also published elsewhere [61].

(Table 1). For incidence risk in the under-fives, however, there was an increase in 10% of districts (mainly around the northern half and easternmost border region), a decrease in 20% of districts around southernmost areas. In comparison, 45% remain unchanged (clustered mainly around the middle half of the country). The mortality trend among the over-fives is more varied (Table 1) with 3%, 69% & 28% of districts either increasing, decreasing or no-change, respectively with a model classification certainty of 75–100% (Fig 5A [ii]).

A large cluster of districts in the southern region has a decreasing trend relative to the rest of the Country (Fig 5B (i-iii). The trend for over-five incidence in Table 1 shows that over half

	Mortality			Incidence		
Age group	Districts	%	Trend	Districts	%	Trend
under 5	19	26.4%	No change	45	62.5%	No change
	51	70.8%	Decrease	20	27.8%	Decrease
	2	2.8%	Increase	7	9.7%	Increase
over 5	20	27.8%	No change	17	23.6%	No change
	50	69.4%	Decrease	10	13.9%	Decrease
	2	2.8%	Increase	45	62.5%	Increase
overall	31	43.1%	No change	34	47.2%	No change
	39	54.2%	Decrease	13	18.1%	Decrease
	2	2.8%	Increase	25	34.7%	Increase

Table 1. Summary description of malaria mortality and incidence trends (<5 years children, \geq 5 years, and all ages combined. NB % represents the proportion of the72 districts assigned to each trend, i.e. Decrease, Increase, or No-change.

https://doi.org/10.1371/journal.pcbi.1008669.t002

of all districts (62%) are increasing, while only 14% are decreasing and 24% exhibit no-change (all results are statistically significant at a 95% credible interval).

A classification matrix for determining overall malaria burden

While rate and risk trend clusters show a clear picture of overall district-level classification, i.e. decline, no-change or increase as illustrated in Fig 4 and Table 1, reviewing these trends separately may conceal or mask the overall underlying picture which in turn might undermine the actual implications of these trends for malaria control. For instance, a district with high risk, high rate, and showing no-change in trend could be more alarming compared to a district that has low risk, low rate and no change or an increasing trend. Therefore, we created a matrix of the combined indices for malaria RIsk, RAtes, and Trends (RIRAT) to accurately classify high-burden and low-burden districts (Fig 6) (*See also S1 Appendix*).

Fig 7 shows comparative district level maps for mortality and incidence burden for the two age categories on the spectrum of *low (green), medium (yellow)*, and *high (red)* -burden (*See <u>S1 Table</u> for details*). Fig 7A and 7C shows the districts in 2015 (mostly in Eastern and Luapula provinces) with the highest under 5 mortality-burden (8 districts) or highest incidence-burden (8 districts) representing an estimated half a million children in that age cohort. Twelve unique districts were classified with either high-mortality or high-incidence burden while four had both. For the overfives, 15 districts were identified as high-incidence burden areas representing approximately 2 million people in that age group. Only two districts had both high-mortality and high incidence burden representing about a quarter-million vulnerable people, while an additional 1.5 million people aged over five lived in the 13 districts with a high-incidence burden only.

Derived from the matrix score, more than 3 million people live in districts with generally high incidence risk, high incidence rates and an increasing trend. This population is exposed to at least twice the risk of malaria compared to other areas in the country.

To assess our method further, we observed differences among incidence classification of those aged \geq 5 years through a comparison of the derived results using raw rates alone against our method. There was a considerable difference in the proportion of districts classified as high or low burden compared to those identified by our method. For example, only 55% of districts identified as high incidence using raw rates alone were also deemed high-burden using the overall weighted combined method. The differences observed here were because high burden districts were showing an overall increasing trend over the period while the others did not. Similarly, 45% of those districts identified as high in the raw rates dropped into the moderate burden category, and 30% of districts identified as low ended up as moderate-burden districts.



Fig 6. Data preparation and processing to determine areas of highest/lowest burden. Stages of data analysis from initial, intermediate, to final outputs. The classes relate to 1, 2, 3 scores with 1 = low, 2 = medium and 3 = high applied to risk, rates, and trends.

The differences observed here highlight the limitations of using raw incidence rates as a basis for identifying and targeting intervention strategies at the subnational level.

Discussion

Our study findings have important implications for malaria policy in Zambia, and the various intervention approaches used within the country. As shown, in both age groups, it is clear that there has been remarkable progress in mortality reduction but less so in incidence reduction. Both the under-five and the over five age groups experienced a similar rate of reduction in mortality (85% and 90% respectively). However, the under-fives continue to experience



Fig 7. High/low burden malaria mortality (A) & (B) and incidence (C) & (D) districts using matrix scores of risk, rate, and trends. https://doi.org/10.1371/journal.pcbi.1008669.g007

approximately five times the incidence rates and at least 2.5 times the mortality burden compared to the over-fives. Without overemphasising the observed declining malaria mortality, the overall results would indicate that more can still be done to further reduce the under-five mortality burden by targeting the highest-burden areas. The benefit of the high precision districtlevel analysis presented in this study provides an opportunity to move away from the *one-sizefits-all* approach, and optimise resource deployment in a more focused, efficient and geographically targeted manner. The findings also demonstrate how a small number of high burden areas can skew the national averages and overshadow the actual progress achieved so far in the country as a whole. This study has provided a means of determining districts with high malaria burden where, if prioritised, targeted malaria control efforts could help maximise impact (*S2 Fig*).

With proposed sub-national elimination approaches soon to be implemented in Zambia, our method, based on an analysis of 16 years of data has identified those areas that are most suitable for malaria elimination (*S2 Fig*). Our method can be applied to help other countries identify high-burden areas and achieve maximum impact through the appropriate use of tools and interventions efficiently and effectively. This is important when considering the use of expensive interventions such as indoor residual spraying, which requires rounds of minimum spray coverage thresholds of up to 85% [46].

An additional point of particular interest in Zambia is that most high-burden areas comprise districts along the national borders with Angola, Democratic Republic of Congo, Malawi, Tanzania, and Mozambique. All these countries are unequivocal, high-burden malariaendemic countries that have often been ranked among the top ten high-burden countries in the world. This observation highlights the significance of the need for countries to engage in bilateral and collaborative regional malaria initiatives for successful control along borders. While Zambia is part of the Elimination8 countries cross-border malaria collaborations, this only applies in southern bordering countries. No such formal undertakings are present with Zambia's northern bordering countries [20]. The patterns and trends presented here reflect Zambia's geographic location and adjacency with contrasting high-burden and low-burden neighbouring countries and highlight the potential influence and impact of cross-border malaria risk in border districts [47,48]. This method, if carefully applied, could additionally benefit other low resource countries and encourage broader regional collaborations, particularly for targeted cross-border initiatives.

We have presented an empirical but rigorous approach for the identification of high-burden/low-burden malaria incidence and mortality in affected countries. In the case of Zambia, we would propose a review of the current under-five malaria intervention strategies, especially in high-mortality burden districts so that any potential problems or issues can be identified and addressed. We would also recommend more focused ongoing operational research to assess progress and identify specific challenges at the community level [49].

While this study focused more on the identification of high burden malaria control areas rather than those most suitable for elimination, the approach still provides sufficient evidence and information that can accurately inform both control and elimination approaches. Our approach provides the information base needed to facilitate further research into the specific factors that might explain within-country differences between regions and age cohorts, including the value and impact of intervention programmes over time. For example, Fig 8 shows the



Fig 8. Incidence and mortality trend against major malaria policy changes. Significant malaria policy changes and guidelines, interventions and diagnostics in Zambia 2000–2015. Note that we did not include any changes that were progressive, e.g. IRS. Major policy changes undertaken in Zambia, 2000–2015 (Source of data: Steketee *et al.* 2008, Chanda *et al.*, 2013, Redditt *et al.* 2012, Kamuliwo *et al.* 2015) [2003: Chloroquine (CQ) replaced by artemisinin-lumefantrine (Coartem) as first-line malaria treatment and new diagnosis and treatment guidelines for malaria to reflect drug policy change launched; 2006: Use of Insecticide Treated Nets (ITNs) adopted; 2006–8: Training of additional microscopists, scale-up of RDTs distribution; free distribution of ITNs through antenatal care (ANC) and intermittent preventive treatment (IPT) using sulfadoxine-pyrimethamine (SP); 2011: Consideration for future elimination begins with the alignment of NMCP strategic plan 2011–2015 with the national vision "*a malaria-free Zambia by* 2030"].

https://doi.org/10.1371/journal.pcbi.1008669.g008

relationship between mortality and incidence trends with significant malaria policy changes and guidelines on interventions and diagnostics in Zambia between 2000 and 2015.

Of interest is the post-2008 trend of increasing incidence rates despite the various intervention strategies. Potential explanations for the observed trends and incidence rate patterns or changing risk patterns include among others low intervention coverage (particularly IRS), insecticide resistance in mosquito nets or low utilisation driven by misuse such as fishing or agriculture fencing or even changing climatic and ecological conditions [31,50]. Our spatiotemporal modelling and the identification of those specific areas where incidence burden and risk is highest provide essential information to support future geographically targeted initiatives. Such initiatives could replace expensive countrywide programmes, thus facilitating more efficient and effective use of scarce resources.

It is recognised that some of the changes in malaria policy, diagnostics, definition, and collection methods during the 16 years may have introduced potential biases in this study. The incorporation of malaria cases by clinical symptoms added some level of non-malarial fever burden, and which could lead to an over-estimation, especially between 2000 and 2008. This bias, however, would be declining in the post-2008 period (*see S1 Appendix*).

For example, the observed changes in prevalence rates over time may in some parts have been influenced by changes in diagnostic tools or methods used in case reporting rather than representing real changes in malaria incidence. This could be more relevant in rural health facility settings [28] where limited availability of trained human resources still exist. We also note that this study gives a long-term time-series of mean trends, risks and rates up to 2015, and therefore presents conclusions accurate for this period of study. Usage for present and future decision-making would have to be based on an analysis of more recent and relatively short time datasets of 3 to 5 years.

Progress in health care provision that could have also impacted some of the identified changes in malaria rates may include the creation of new health facilities and the introduction of Community Health Workers (CHWs), particularly since 2010. The assumption is that both these factors could increase access to health care services which in turn would artifactually result in higher treatment-seeking rates and higher case diagnoses. The increase in CHWs is supported by evidence from the MIS 2018 [51] which shows that CHW activity increased during the 2010 to 2015 period of our study. During that period, malaria medication received from CHWs increased across previous MIS surveys from 2% in 2010, 8% in 2012 to 25% in 2015 but down slightly to 22% in 2018 (see Fig 9 and S4 Fig). This increase could suggest that CHWs are a possible cause or contributor to the increase in incidence in some areas. However, this argument would be further supported if the increase in CHW treatments was matched with a similar increase in treatment-seeking behaviour. In fact, national data trends show a considerable reduction in treatment-seeking behaviour during the same period in both rural and urban areas (see Fig 9 and S4 Fig). So while nationally, there has been an increase in the proportion of patients being treated by CHWs through increased numbers and availability, the actual total proportions of those people seeking treatment has decreased.

This also relates to the assumption that increasing numbers of health facilities may explain some of the increasing incidence trends. Zambia's primary reason for the construction of new health facilities is often based on population growth with catchment distance and accessibility issues a secondary factor [52–55]. During the period of this study from 2009 to 2015, the number of health facilities reporting in the HMIS increased from 1552 in 2009 to 1869 in 2015. The increase was made up of a combination of government, mission (faith-based), and private providers [52]. During the same period, Zambia's population increased by 18.5%, while the total number of health facilities increased by 18.7%, and the ratio of health facilities to population remained the same throughout (1.2 health facilities per 10 000 population).



*T.S stands for Treatment-seeking.

Fig 9. Treatment-Seeking behaviours and Community Health Worker medications in Zambia 2006–2018. * T.S stands for Treatment-seeking. Data source: MIS reports 2006, 2008, 2010, 2015, & 2018.

https://doi.org/10.1371/journal.pcbi.1008669.g009

Based on a comparison between population growth and the increase in health facilities, it seems obvious and acceptable to suggest that the construction of new health facilities was primarily driven by population growth. It can also be further argued that the population growth ratio was consistent with the provision of health care facilities per person. Similarly, the incidence rates of malaria would also have stayed the same unless other factors were at play. It would be expected that in either scenario malaria incidence rates would be normalised by the relevant underlying catchment populations and therefore would not likely increase incidence significantly unless the introduction of new facilities resulted in increased treatment-seeking rates. We have seen from Fig 9 that the national treatment-seeking rates in the country actually fell as reported across all the MISs in nearly all provinces and across the rural-urban divide [51,56–60]. In addition, an analysis of national Health Facility data by District from 2009–2015 found a small, but significant, negative correlation (-0.16) between the number of health facilities and incidence rates with over 50% of districts that had an increase in health facility provision actually showing a declining malaria trend and only 33% showing an increase.

In this study, other potential limitations that could have influenced some of the observed results may include: i) the potential presence of unquantifiable effects due to the lack of reliable subnational treatment-seeking rates capable of indicating existing subnational variations if present, ii) uncaptured subclinical malaria which is long known to have a severe impact on transmission, especially in the older age groups due to partial immunity; and iii) the unknown effects of any differences on how quickly RDT use was adopted across the country. These, if present, may affect our conclusions of what the malaria burden in the population actually is (*see S1 Appendix for details on asymptomatic malaria*).

Having said that, the increasing availability (reduced lag) and improvement (in accuracy) of health management information system (HMIS) data presented here provide a much

greater opportunity for such data to be used with more confidence in the future. This is particularly true given the more expensive alternatives such as surveys that may not always provide comprehensive longitudinal information and analysis at the times when it is most needed.

Conclusion

We have presented a method here that augments conventional measures of identifying malaria risk and provides an effective approach for the identification of areas of high and low malaria burden at the sub-national level within countries. By applying a rigorous spatio-temporal approach that uses longitudinal rates, risks and trend clusters, we can help policymakers determine priority areas to deploy scarce resources for high impact control interventions in high-burden areas and elimination strategies in low burden areas.

This easy to implement and replicate methodology will help policy makers and malaria control/elimination program staff in malaria-endemic countries who may not be fully cognisant of or technically skilled in advanced statistical methods (See <u>S1 Code</u>). The novelty of our method is not in the statistical algorithms, which are well-established techniques in their own right, but in the approach of combining the typically independent measures of rates, risk, and trend over time and space that better represent malaria prevalence within a country and are easy to replicate and use at an operational and practical planning level.

We believe that applying this approach could be extremely beneficial to countries embarking on their malaria elimination strategies as part of the global malaria eradication agenda. This could be particularly effective through informed sub-national programs at even finer levels of geographic aggregation, such as health facility catchments, which are well suited for targeted control and elimination strategies.

Supporting information

S1 Fig. Average temporal trends at 95% credible intervals. *Estimated temporal trends and 95% credible intervals in dotted lines, arranged to start with under-five (i), over-five (ii) and population-wide (iii), for malaria mortality (A), and malaria infections (B).* (TIF)

S2 Fig. Maps showing districts with the highest and lowest scores only based on all matrix scores. [Trend u5, Risk u5, Rate u5, Trend o5, Risk o5 & Rate o5] (TIFF)

S3 Fig. HMIS data Quality improvements from 2010–2016. (TIF)

S4 Fig. Prompt Treatment-seeking rates, 2006–2018. *Treatment-seeking trends between rural and urban areas of Zambia during the period of study.* (TIF)

S1 Table. Arbitrary combinations of score legends, colours and their resultant score and colour.

(XLSX)

S1 Appendix. Additional text and explanation of methods. (DOCX)

S1 Code. R code used for part of the analysis upon which the manuscript is based. (R)

Acknowledgments

We are grateful for the help rendered by Gary Napier of the University of Glasgow for checking that the use of his trends R code worked as intended.

Author Contributions

Conceptualization: Jailos Lubinda, Adrian J. Moore.

Data curation: Jailos Lubinda, Busiku Hamainza.

Formal analysis: Jailos Lubinda, Ubydul Haque.

Funding acquisition: Yaxin Bi, Busiku Hamainza, Ubydul Haque, Adrian J. Moore.

Investigation: Ubydul Haque, Adrian J. Moore.

Methodology: Jailos Lubinda, Yaxin Bi, Ubydul Haque, Adrian J. Moore.

Project administration: Jailos Lubinda, Busiku Hamainza, Adrian J. Moore.

Resources: Busiku Hamainza.

Software: Jailos Lubinda, Ubydul Haque.

Supervision: Yaxin Bi, Busiku Hamainza, Ubydul Haque, Adrian J. Moore.

Validation: Yaxin Bi, Busiku Hamainza.

Visualization: Jailos Lubinda.

Writing - original draft: Jailos Lubinda, Ubydul Haque, Adrian J. Moore.

Writing - review & editing: Yaxin Bi, Ubydul Haque, Adrian J. Moore.

References

- 1. World Health Organization. World Malaria Report 2018 [Internet]. World Health Organization: Geneva; 2018. Available from: https://www.who.int/malaria/media/world-malaria-report-2018/en/.
- 2. World Health Organization. World Malaria Report 2017. World Health Organization: Geneva; 2017.
- Alonso P, Noor AM. The global fight against malaria is at crossroads. Lancet [Internet]. World Health Organization. Published by Elsevier Ltd/Inc/BV. All rights reserved.; 2017; 6736:1–3. Available from: http://linkinghub.elsevier.com/retrieve/pii/S0140673617330805.
- World Health Organization, RBM. High burden to high impact A targeted malaria response advance copy. 2018;1–8. Available from: http://apps.who.int/iris/bitstream/handle/10665/275868/WHO-CDS-GMP-2018.25-eng.pdf?ua = 1.
- Ghebreyesus TA, Admasu K. Countries must steer new response to turn the malaria tide. Lancet [Internet]. World Health Organization. Published by Elsevier Ltd/Inc/BV. All rights reserved.; 2018; 0:2246– 7. Available from: https://linkinghub.elsevier.com/retrieve/pii/S014067361832943X.
- National Malaria Elimination Programme. National Malaria Elimination Strategic Plan 2017–2021: Moving from accelerated burden reduction to malaria elimination in Zambia. 2017;1–43. Available from: http://www.makingmalariahistory.org/wp-content/uploads/2017/06/NMESP_2017-21_email-version. pdf.
- Kamuliwo M, Chanda E, Haque U, Mwanza-Ingwe M, Sikaala C, Katebe-Sakala C, et al. The changing burden of malaria and association with vector control interventions in Zambia using district-level surveillance data, 2006–2011. Malar J. 2013; 12:1–9. <u>https://doi.org/10.1186/1475-2875-12-1</u> PMID: 23282136
- Masaninga F, Chanda E, Chanda-Kapata P, Hamainza B, Masendu HT, Kamuliwo M, et al. Review of the malaria epidemiology and trends in Zambia. Asian Pac J Trop Biomed. 2013; 3:89–94. <u>https://doi.org/10.1016/S2221-1691(13)60030-1</u> PMID: 23593585
- Chizema-Kawesha E, Miller JM, Steketee RW, Mukonka VM, Mukuka C, Mohamed AD, et al. Scaling up malaria control in Zambia: Progress and impact 2005–2008. Am J Trop Med Hyg. 2010; 83:480–8. https://doi.org/10.4269/ajtmh.2010.10-0035 PMID: 20810807

- National Malaria Control Programme. National Malaria Control Programme Strategic Plan For FY 2011–2015. 2012;
- Comfort A, Leegwater A, Nakhimovsky S, Kansembe H, Hamainza B, Bwalya B, et al. Exploring the use of routinely-available, retrospective data to study the association between malaria control scale-up and micro-economic outcomes in Zambia. Malar J. BioMed Central; 2017; 16:1–19.
- 12. E8 Secretariate. State of Malaria in the E8 [Internet]. 2018 [cited 2018 Dec 20]. Available from: https://malariaelimination8.org/state-malaria-e8/.
- 13. President's Malaria Initiative (PMI). Malaria Operational Plan FY 2017 Zambia. 2017.
- Zambia National Malaria Elimination Centre. National Malaria Elimination Strategic Plan 2017–2021: Moving from accelerated burden reduction to malaria elimination in Zambia. 2017; Available from: http://www.makingmalariahistory.org/wp-content/uploads/2017/06/NMESP_2017-21_email-version. pdf.
- Loewenberg S. Zambia's drive to eliminate malaria faces challenges. Bull World Health Organ. World Health Organization; 2018; 96:302–3. https://doi.org/10.2471/BLT.18.020518 PMID: 29875513
- 16. Kitojo C, Gutman JR, Chacky F, Kigadye E, Mkude S, Mandike R, et al. Estimating malaria burden among pregnant women using data from antenatal care centres in Tanzania: a population-based study. Lancet Glob Heal [Internet]. The Author(s). Published by Elsevier Ltd. This is an Open Access article under the CC BY-NC-ND 4.0 license; 2019; 7:e1695–705. Available from: http://dx.doi.org/10.1016/ S2214-109X(19)30405-X.
- 17. Routledge I, Lai S, Battle KE, Ghani AC, Gomez-Rodriguez M, Gustafson KB, et al. Tracking progress towards malaria elimination in China: estimates of reproduction numbers and their spatiotemporal variation. bioRxiv. Cold Spring Harbor Laboratory; 2019;628842.
- 18. Amratia E., Cameron A, Young K, Twohig A, Python D, Bhavani E, et al. Mapping for malaria control in Grand'Anse, Haiti using a multi-metric Bayesian approach. ASTMH 68th Annu Meet. Gaylord National Resort and Convention Center National Harbor, Maryland USA; 2019.
- Central Statistical Office. 2010 census of population and housing: Population and Demographic Projections 2011–2035. 2013;199. Available from: <u>http://www.zamstats.gov.zm/phocadownload/Zambia</u> Census Projection 2011–2035.pdf.
- Elimination8 Secretariat. Working towards a malaria-free Southern Africa [Internet]. 2017 [cited 2017 Jan 2]. Available from: http://malariaelimination8.org/.
- 21. Central Statistical Office. 2010 Zambia Census of Population and Housing: Summary Report. 2012;
- Murray CJL, Rosenfeld LC, Lim SS, Andrews KG, Foreman KJ, Haring D, et al. Global malaria mortality between 1980 and 2010: a systematic analysis. Lancet. Elsevier; 2012; 379:413–31. <u>https://doi.org/10.1016/S0140-6736(12)60034-8 PMID: 22305225</u>
- 23. World Health Organization. World Malaria Report 2008. World Health Organization: Geneva; 2008.
- 24. World Health Organization. World Malaria Report 2011. World Health Organization: Geneva; 2011.
- Yukich JO, Bennett A, Albertini A, Incardona S, Moonga H, Chisha Z, et al. Reductions in artemisininbased combination therapy consumption after the nationwide scale up of routine malaria rapid diagnostic testing in Zambia. Am J Trop Med Hyg. 2012; 87:437–46. https://doi.org/10.4269/ajtmh.2012.12-0127 PMID: 22848096
- 26. Chanda E, Kamuliwo M, Steketee RW, Macdonald MB, Babaniyi O, Mukonka VM. An overview of the malaria control programme in Zambia. ISRN Prev Med [Internet]. 2013; 2013:495037. Available from: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid = 4062855&tool = pmcentrez&rendertype = abstract. https://doi.org/10.5402/2013/495037 PMID: 24967138
- Steketee RW, Sipilanyambe N, Chimumbwa J, Banda JJ, Mohamed A, Miller J, et al. National malaria control and scaling up for impact: The Zambia experience through 2006. Am J Trop Med Hyg. 2008; 79:45–52. PMID: 18606763
- Mukonka V., Chanda E, Kamuliwo M, Elbadry MA, Wamulume PK, Mwanza-Ingwe M, et al. Diagnostic approaches to malaria in Zambia, 2009–2014. Geospat Health. 2015; 10:330. https://doi.org/10.4081/ gh.2015.330 PMID: 26054519
- Reid HL, Haque U, Roy S, Islam N, Clements AC. Characterizing the spatial and temporal variation of malaria incidence in Bangladesh, 2007. Malar J [Internet]. 2012; 11:170. Available from: Malaria Journal. https://doi.org/10.1186/1475-2875-11-170 PMID: 22607348
- 30. Mabaso MLH, Vounatsou P, Midzi S, Da Silva J, Smith T. Spatio-temporal analysis of the role of climate in inter-annual variation of malaria incidence in Zimbabwe. Int J Health Geogr [Internet]. 2006; 5:20. Available from: http://www.scopus.com/inward/record.url?eid = 2-s2.0-33746357890&partnerID = tZOtx3y1. https://doi.org/10.1186/1476-072X-5-20 PMID: 16700905
- Bennett A. Determinants of the spatial and temporal distribution of malaria in Zambia and association with vector control. ProQuest Diss Theses [Internet]. 2012;166. Available from: http://sfx.scholarsportal.

info/guelph/docview/1610814006?accountid = 11233%5Cnhttp://sfx.scholarsportal.info/guelph?url_ver = Z39.88-2004&rft_val_fmt = info:ofi/fmt:kev:mtx:dissertation&genre = dissertations+%26+theses&sid = ProQ:ProQuest+Dissertations+%26+Theses+A.

- Hamra G, MacLehose R, Richardson D. Markov chain monte carlo: An introduction for epidemiologists. Int J Epidemiol. 2013; 42:627–34. https://doi.org/10.1093/ije/dyt043 PMID: 23569196
- Lee D, Rushworth A, Napier G. Spatio-Temporal Areal Unit Modeling in R with Conditional Autoregressive Priors Using the CARBayesST Package. J Stat Softw [Internet]. 2018; 84. Available from: http://www.jstatsoft.org/v84/i09/.
- Charras-Garrido M, Abrial D, Goër J De, Dachian S, Peyrard N. Classification method for disease risk mapping based on discrete hidden Markov random fields. Biostatistics [Internet]. 2012; 13:241–55. Available from: https://doi.org/10.1093/biostatistics/kxr043 PMID: 22133757
- Napier G, Lee D, Robertson C, Lawson A. A Bayesian space-time model for clustering areal units based on their disease trends. Biostatistics [Internet]. 2018; 00:1–17. Available from: https://academic. oup.com/biostatistics/advance-article-abstract/doi/10.1093/biostatistics/kxy024/5039880.
- **36.** R Core Team. R: A Language and Environment for Statistical Computing: R Foundation for Statistical Computing. 2013; Available from: http://www.r-project.org/.
- R Core Team. R Core Team (2018). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. http://www. R-project. org;
- Napier G, Lee D, Robertson C, Lawson A, Pollock KG. A model to estimate the impact of changes in MMR vaccine uptake on inequalities in measles susceptibility in Scotland. Stat Methods Med Res [Internet]. SAGE Publications Ltd STM; 2016; 25:1185–200. Available from: <u>https://doi.org/10.1177/</u> 0962280216660420 PMID: 27566772
- **39.** World Bank. Zambia Study Shows Stronger Supply Chains for Key Drugs Can Reduce Child Mortality. Washington, DC; 2010.
- Zambia Ministry of Health Logistics Pilot Program Steering Committee. Essential Medicines Logistics Pilot Program: Steering Committee Evaluation Report. Lusaka; 2011.
- **41.** Yadav P. Improving public health in developing countries through operations research. Wiley Encycl Oper Res Manag Sci. Wiley Online Library; 2010;
- 42. Leung N-HZ, Chen A, Yadav P, Gallien J. The impact of inventory management on stock-outs of essential drugs in Sub-Saharan Africa: secondary analysis of a field experiment in Zambia. PLoS One. Public Library of Science; 2016; 11:e0156026. https://doi.org/10.1371/journal.pone.0156026 PMID: 27227412
- **43.** USAID | Deliver Project TO 4 and 7. USAID | Deliver Project Final Country Report: Zambia. Arlington; 2016.
- 44. President's Malaria Initiative. President's Malaria Initiative Zambia Malaria Operational Plan FY 2019. 2019; Available from: https://www.pmi.gov/docs/default-source/default-document-library/malaria-operational-plans/fy19/fy-2019-kenya-malaria-operational-plan.pdf?sfvrsn = 3.
- Vledder M, Friedman J, Sjöblom M, Brown T, Yadav P. Improving Supply Chain for Essential Drugs in Low-Income Countries: Results from a Large Scale Randomized Experiment in Zambia. Heal Syst Reform [Internet]. Taylor & Francis; 2019; 5:158–77. Available from: https://doi.org/10.1080/23288604. 2019.1596050.
- 46. Pinder M, Jawara M, Jarju LBS, Salami K, Jeffries D, Adiamoh M, et al. Efficacy of indoor residual spraying with dichlorodiphenyltrichloroethane against malaria in Gambian communities with high usage of long-lasting insecticidal mosquito nets: a cluster-randomised controlled trial. Lancet. Elsevier; 2015; 385:1436–46. https://doi.org/10.1016/S0140-6736(14)61007-2 PMID: 25498847
- Simon C, Moakofhi K, Mosweunyane T, Jibril HB, Nkomo B, Motlaleng M, et al. Malaria control in Botswana, 2008–2012: the path towards elimination. Malar J. BioMed Central; 2013; 12:458. <u>https:// doi.org/10.1186/1475-2875-12-458 PMID: 24359262</u>
- Chihanga S, Haque U, Chanda E, Mosweunyane T, Moakofhi K, Jibril HB, et al. Malaria elimination in Botswana, 2012–2014: achievements and challenges. Parasit Vectors. BioMed Central; 2016; 9:99. https://doi.org/10.1186/s13071-016-1382-z PMID: 26911433
- Haque U, Hashizume M, Sunahara T, Hossain S, Ahmed SM, Haque R, et al. Progress and challenges to control malaria in a remote area of Chittagong hill tracts, Bangladesh. Malar J. BioMed Central; 2010; 9:156.
- Bennett A, Yukich J, Miller JM, Keating J, Moonga H, Hamainza B, et al. The relative contribution of climate variability and vector control coverage to changes in malaria parasite prevalence in Zambia 2006–2012. Parasit Vectors [Internet]. Parasites & Vectors; 2016; 9:431. Available from: http://parasitesandvectors.biomedcentral.com/articles/10.1186/s13071-016-1693-0.
- 51. Zambia Ministry of Health. Zambia Malaria Indicator Survey-MIS 2018. 2019.

- 52. Zambian Ministry of Health. The 2012 List of Health Facilities in Zambia: Preliminary Report, v15. Ministry of Health Lusaka; 2013.
- Ferrinho P, Siziya S, Goma F, Dussault G. The human resource for health situation in Zambia: Deficit and maldistribution. Hum Resour Health. 2011; <u>https://doi.org/10.1186/1478-4491-9-30</u> PMID: 22182366
- Hoppenbrouwer J, Kanyengo CW. Current access to health information in Zambia: A survey of selected health institutions. Health Info Libr J. 2007; <u>https://doi.org/10.1111/j.1471-1842.2007.00728.x</u> PMID: 18005299
- 55. CSO. 2010 Census of population and housing: National Analytical Report. Natl Anal Rep. 2012;
- 56. Zambian Ministry of Health. Zambia National Malaria Indicator Survey 2006. 2006;38-41.
- 57. Zambian Ministry of Health. The Zambia National Malaria Indicator Survey 2008. 2008; Available from: file:///C:/Users/youhe/Downloads/kdoc_o_00042_01.pdf.
- Zambian Ministry of Health. Zambia Malaria Indicator Survey 2015 [Internet]. 2015. Available from: http://www.makingmalariahistory.org/wp-content/uploads/2017/06/Zambia-MIS2015_Jan20-nosigs. pdf.
- 59. Zambian Ministry of Health. Zambia National Malaria Indicator Survey 2012. 2012; Available from: file:///C:/Users/youhe/Downloads/kdoc_o_00042_01.pdf.
- Zambian Ministry of Health. Zambia National Malaria Indicator Survey 2010. <u>Malariasurveys.org</u> [Internet]. 2010; Available from: http://data.worldbank.org/indicator/AG.PRD.FOOD.XD.
- Lubinda J, Haque U, Bi Y, Hamainza B, Moore AJ. Near-term climate change impacts on sub-national malaria transmission. Sci Rep. Nature Publishing Group; 2021; 11:1–13. <u>https://doi.org/10.1038/s41598-020-79139-8 PMID</u>: 33414495